Author's response to reviews

Title: Obliteration of radical cavities with autogenous cortical bone; long-term results

Authors:

Akram M Abdelrahman (akram.abdelrahman@gmail.com)
Matti Pietola (matti.pietola@hus.fi)
Teemu J Kinnari (teemu.j.kinnari@helsinki.fi)
Hans Ramsay (hans.ramsay@hus.fi)
Jussi Jero (jussi.jero@hus.fi)
Antti A Aarnisalo (antti.aarnisalo@helsinki.fi)

Version: 2 Date: 24 May 2008

Author's response to reviews:

Attached please find our revised manuscript Obliteration of radical cavities with autogenous cortical bone; long-term results by authors AM Abdel-Rahman, M Pietola, TJ Kinnari, H Ramsay, J Jero and AA Aarnisalo.

We have answered in detail to all the questions and comments of the three reviewers.

Our answers are attached to this resubmission. We hope that the manuscript could now be accepted for publication in BMC Ear Nose and Throat Disorders.

Looking forward to hearing you soon,
Sincerely yours,
Antti A. Aarnisalo,MD